

Delft University of Technology

Frameworks for multi-system innovation analysis from a sociotechnical perspective A systematic literature review

Bakhuis, Jerico; Kamp, Linda Manon; Barbour, Natalia; Chappin, Émile Jean Louis

DOI 10.1016/j.techfore.2024.123266

Publication date 2024 Document Version Final published version

Published in Technological Forecasting and Social Change

Citation (APA)

Bakhuis, J., Kamp, L. M., Barbour, N., & Chappin, É. J. L. (2024). Frameworks for multi-system innovation analysis from a sociotechnical perspective: A systematic literature review. *Technological Forecasting and Social Change*, *201*, Article 123266. https://doi.org/10.1016/j.techfore.2024.123266

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.



Contents lists available at ScienceDirect

Technological Forecasting & Social Change

journal homepage: www.elsevier.com/locate/techfore



Frameworks for multi-system innovation analysis from a sociotechnical perspective: A systematic literature review



Jerico Bakhuis^{a,*}, Linda Manon Kamp^a, Natalia Barbour^b, Émile Jean Louis Chappin^a

^a Delft University of Technology, Faculty of Technology, Policy & Management, Jaffalaan 5, NL-2628 BX Delft, The Netherlands
 ^b University of Central Florida, Department of Civil, Environmental and Construction Engineering, 12800 Pegasus Dr #211, Orlando, FL 32816, United States

ARTICLE INFO

Keywords: Sustainability transition Multi-system Sector coupling Converging infrastructures Innovation systems Multi-level perspective

ABSTRACT

This paper systematically reviews the literature on sociotechnical multi-system innovation frameworks that broaden the usual focus on one sociotechnical system to encompass influences from multiple systems. The review includes 75 peer-reviewed papers that span a broad range of energy-demanding systems and mainly build upon the core frameworks of the Multi-level Perspective (MLP) and Technological Innovation Systems (TIS). The analysis identifies three key aspects to consider in multi-system frameworks. The first aspect is the importance of considering the overarching directionality of multiple sociotechnical systems and how they influence each other. The second is to explicitly analyse the phase of each transitioning system. The third aspect is a need for explicit system configuration analysis. This includes analysing the value chain and the number and types of sectors linked to it, typifying the distinct characteristics of sectors internally and how they interact, and analysing complementary or competitive technologies. The paper concludes by providing recommendations for future research, with a particular focus on the further development of new multi-system frameworks that include one or more of the prior-mentioned three key takeaways. Firstly, focusing on dynamics within multi-system niches. Secondly, performing actor-level analysis, including demand-side analysis. Finally, applying quantitative methods, such as computer simulation modelling.

1. Introduction

This paper stems from the increased need for multi-system innovations in the current phase of the energy transition, as the ambitious climate goals to limit global warming to well below 2 degrees Celsius (UNFCCC, 2015) require low-carbon transitions across multiple sociotechnical systems. In this context, sociotechnical systems refer to linkages between a large variety of social and physical elements necessary to fulfil societal functions (e.g., electricity, mobility, or heating) (Geels, 2004; Markard, 2011).

In line with the climate goals, the current 'energy transition' to decrease fossil-based energy supply to stay within the carbon emissions budget (Rogelj et al., 2019) encompasses transitions of multiple (interacting) sociotechnical systems that need to decrease carbon emissions. As the energy transition progresses, next to transformations of individual sociotechnical systems, the necessity of interactions between sociotechnical systems becomes increasingly critical (Robinius et al., 2017a; Geels et al., 2017a). For instance, in order to efficiently integrate the increasing intermittent renewable electricity sources (such as wind and the sun) into the electricity system, a coupling of the electricity sector to other energy-demanding sectors such as transport, industry and the built environment is needed – generally referred to as 'sector coupling' (Fridgen et al., 2020; Brown et al., 2018). Such system linkages can also positively affect the momentum of sociotechnical transitions (Geels, 2018). For example, Vehicle-to-Grid (V2G) configurations, in which electric vehicles can return electricity to the grid when necessary, can promote the adoption of electric vehicles and help mitigate the intermittency problems of wind and solar electricity (Sovacool et al., 2017). The need for such solutions is becoming more pressing as, in the current phase of the energy transition, intermittent renewable electricity is increasingly diffusing in electricity grids (e.g., causing congestion issues) (Markard, 2018; Mitchell, 2016; Turnheim et al., 2018).

Therefore, there is an increasing necessity to study the interplay of multiple systems along with multi-system innovations (e.g., V2G or intelligent heating systems) in the current phase of the energy transition. In this paper, we use the broad term 'multi-system' to encompass the

* Corresponding author.

https://doi.org/10.1016/j.techfore.2024.123266

Received 24 May 2023; Received in revised form 3 November 2023; Accepted 3 February 2024 Available online 15 February 2024

0040-1625/© 2024 The Author(s). Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

E-mail addresses: J.J.Bakhuis-1@tudelft.nl (J. Bakhuis), L.M.Kamp@tudelft.nl (L.M. Kamp), Natalia.Barbour@ucf.edu (N. Barbour), E.J.L.Chappin@tudelft.nl (É.J.L. Chappin).

influence of multiple sectors¹ (multi-sector) and regimes² (multiregime) on each other. The implementation of multi-system innovations adds a layer of complexity since their uptake depends on increasing integration of two or more heterogeneous sociotechnical systems, which can vary substantially, e.g., in terms of historical precedence (norms and standards), knowledge base, actors (types and motivations), technologies, inputs, demands and culture (e.g., Malerba, 2002; Patel and Pavitt, 1994; Dolata, 2009). For example, the success of V2G configurations depends on many social and technical developments across the electricity and road transport systems. A survey of the scientific literature about multi-system innovations such as heat pumps (e.g., Böttger et al., 2014; Lund et al., 2010), synthetic fuels (e.g., biofuels or hydrogen) and electric cars (e.g., Lund and Kempton, 2008; Mwasilu et al., 2014), as well as the emerging sector coupling literature (e.g., Bloess, 2019; Child et al., 2018; Ramsebner et al., 2021), shows that these innovations are mainly studied from a technical or economic perspective. Therefore, there is a need to employ appropriate sociotechnical frameworks to encompass the full range of important aspects of studying multi-system innovations.

The field of (sustainability) 'transition studies' focuses on the study of complex, long-term, fundamental transformations of sociotechnical systems. These studies adopt a systems perspective, considering the mutual influence of social, technical, institutional, political, economic and socio-cultural aspects, and emphasise path-dependency and lock-in (Schot and Geels, 2008; Markard et al., 2012; Markard, 2018). The frameworks used in these studies have predominantly been developed and are mainly used to analyse the emergence of a focal (niche) technology in one specific sector (Geels, 2018; Markard et al., 2016; Raven, 2007), e.g., natural gas replacing a coal-based regime (Correlje and Verbong, 2004) or steamships replacing the sailing ship regime (Geels, 2002). However, within the literature, there have been several recent calls for expansion to consider broader contextual influences, including multi-system interactions (e.g., Markard, 2018; Schot and Kanger, 2018; Geels et al., 2017a). This has been followed by disparate multi-system framework suggestions and adaptations attempting to (partly) address this gap (e.g., Papachristos et al., 2013; Malhotra et al., 2019).

The current work aims to systematically review 'transition studies' literature (also known as transitions literature) to make an inventory of available analytical frameworks and investigate their applicability to study innovations spanning multiple systems (i.e., multi-system innovations) from a sociotechnical perspective. This paper primarily seeks to answer the following research question: "What insights do papers from the field of transition studies provide for the study of multi-system innovations from a sociotechnical perspective?"

In this review, we start from a broad perspective to include former research that attempted to extend the usual single-system focus to consider multi-system influences. To narrow the search area, we focus on studies that include two (inter-system) or more (multi-system) of the following energy-demanding sociotechnical systems: electricity, transport, industry, the built environment, and agriculture. Within this scope, e.g., studies that explicitly include the sector coupling concept as described above or that focus on multi-system innovations that anchor in two or more of the abovementioned systems are of particular interest. The purpose is to add value through the exploration of existing Technological Forecasting& Social Change 201 (2024) 123266

transition frameworks to be able to study multi-system innovations from a sociotechnical perspective.

This paper is structured as follows. Section 2 provides a brief theoretical background. Section 3 describes the literature review methodology, including the search strategy and evaluation criteria of the included studies. Section 4 contains the results of this work in the form of a descriptive analysis of the multi-system frameworks used in the reviewed papers. Finally, Section 5 provides conclusions of the research results, discusses the key findings from the reviewed studies and articulates a future research agenda.

2. Theoretical background

Transition studies emerged in response to pressing environmental issues, like climate change, recognising the need for transformative change in the form of a complex non-linear transition process (Rip and Kemp, 1998). At its core, it takes a holistic sociotechnical perspective to conceptualise and explain the profound reconfigurations necessary across technology, actors, and institutions (Geels, 2002). To decipher the intricate dynamics of sociotechnical transitions, the field draws underlying concepts from various other fields, including evolutionary economics, social construction of technology, neo-institutional theory and complex adaptive systems theory (Geels, 2004; Geels and Schot, 2007). Based on these concepts, over the past two decades, core frameworks have been developed for comprehending transition processes. Two widely embraced frameworks include the Multi-Level Perspective (MLP) (Geels, 2002; Geels and Schot, 2007; Smith et al., 2010) and Technological Innovation Systems (TIS) (Hekkert et al., 2007; Bergek et al., 2008, 2015; Markard and Truffer, 2008).

The Multi-Level Perspective (MLP) framework traditionally consists of three interdependent analytical levels that each play a major role in contributing to a particular transition or technological innovation's success (Rip and Kemp, 1998; Smith et al., 2010). Central to this framework is the sociotechnical regime, which encompasses the established systems that impose path-dependent incremental sociotechnical change (Geels, 2002). Niches play an important role in the emergence of radical innovations, providing 'protected spaces' from selection pressure imposed by the regime (Hoogma et al., 2002; Smith and Raven, 2012). Finally, the sociotechnical landscape includes influential developments external to the regime and niche (e.g., political, cultural or macroeconomic), which can form an undercurrent and be relatively slow. Regimeinternal or -external tensions exerted by landscape developments can lead to windows of opportunity for niche innovations to gain momentum (Geels et al., 2018). A primary academic focus of this framework is on regime shifts and the factors that lead to the destabilisation of existing regimes and the emergence of new regimes (as illustrated in Fig. 1). Another analytical focus is on gaining deeper understanding of the innovations incubated in niches and evaluating their performance, for example by analysing internal niche dynamics, such as actor expectations, network building and learning processes.³

Within the traditional Technological Innovation System (TIS) framework⁴ (Carlsson and Stankiewicz, 1991; Hekkert et al., 2007; Bergek et al., 2015; Weber and Truffer, 2017), there is an analytical focus on the development, diffusion, and use of a particular technology

 $^{^{1}\ \}mathrm{A}$ sector consists of actors, institutions and technologies that generate products (e.g., cars or electronics) or services (e.g., transport or energy) and may use several technologies (e.g., solar or wind). Where to draw the boundaries between sectors depends on the scope of the analysis, e.g., if the competencies needed for a service differ substantially (e.g., in engineering).

² A (sociotechnical) regime can be understood as the 'deep-structure' or grammar of sociotechnical systems, including technical and social aspects that account for the stability of sociotechnical configurations. This includes engineering practices, production technologies, skills and procedures, and the institutions and infrastructures they are embedded in (Geels, 2002, 2004).

 $^{^{3}\,}$ To understand the internal niche dynamics and evaluate a niche's performance, the Strategic Niche Management (SNM) framework was developed based on similar conceptual grounds (Schot and Geels, 2008; Markard et al., 2012); SNM provides conceptual tools to assess the performance of niche dynamics (e.g., Van der Laak et al., 2007; Hoogma et al., 2002).

The TIS framework has emerged within a broader field of Innovation System (IS) concepts, including national innovation systems (NIS) (Freeman, 1987; Lundvall, 1992), regional innovation systems (RIS) (Asheim and Isaksen, 1997; Cooke et al., 1997) and Sectoral Innovation Systems (SIS) (Breschi and Malerba, 1997; Malerba, 2002).



Fig. 1. Multi-Level Perspective (MLP) framework (Geels et al., 2017b).

from a sociotechnical systems perspective, encompassing all components that influence its innovation process (including actors, institutions, and other technologies) (see Fig. 2). A TIS may be a subsystem of a sectoral system or may cut across several sectors (Markard and Truffer, 2008). From this perspective, the development of a new technology results from the positive fulfilment of seven functions: knowledge development and diffusion, entrepreneurial experimentation, influence on the direction of search, market formation, legitimation, resource mobilisation and development of positive externalities (Bergek et al., 2008). Beyond gaining a deeper understanding of internal niche dynamics, a main focus is on uncovering 'system failures' by mapping these key processes to provide a wide range of practical policy recommendations.

As addressed in the introduction, existing conceptual frameworks primarily focused on singular sociotechnical systems (i.e. sectors or regimes). As the energy transition progresses, there is a growing need for framework adaptations that can address transitions spanning multiple sociotechnical systems (Markard, 2018; Köhler et al., 2019). The following section (Section 3) details the research methodology, after which Section 4 synthesises the systematic review results regarding disparate framework adaptations in the literature addressing this gap.

3. Research methodology

The current work systematically reviews the analytical frameworks in transitions literature to synthesise their contributions towards studying multi-system interactions and associated multi-system innovations from a sociotechnical systems perspective. In order to add rigour and thoroughness for conducting such a systematic literature review, we turn to existing guidelines for reviews focussing on framework and conceptual contributions (Wee and Banister, 2016; Snyder, 2019; Xiao and Watson, 2019). Furthermore, we follow the recommendation of various authors to employ transparency and process analysis documentation (Torraco, 2005; MacInnis, 2011; Snyder, 2019).

Considering the breadth of the research question, the choice was made to focus on one broad, emerging body of literature (i.e., transition studies). This review does not consider literature outside of this field, which could grant insights into the study of multi-system innovations from a sociotechnical perspective (e.g., governance studies). Additionally, due to the broad range of concepts and fast-growing database of case studies, the selection criteria were strict to identify only the key and most relevant papers. Only studies written in English from the transitions literature were considered within this scope. Furthermore, both conceptual and empirical (case) studies were included when there was evidence of a multi-system focus (including multi-sector and multi-regime). Moreover, prior to the systematic search, since the literature contains many technical terms, an initial search was done to identify relevant search terms concerning both the theoretical concepts within the literature and the specific terms encompassing multi-system aspects. See Table A.1 in the appendix for a list of the search terms. Considering this, the systematic search was conducted in four separate steps (identification, screening, eligibility and inclusion), as visualised in Fig. 3.

In the first step (identification), a search was done by combining each of the 'transitions literature terms' with the specific 'multi-system terms' found in the initial search and by using quotation marks and the BOOLEAN operators AND/OR. More specifically, the following search string was employed, incorporating every possible combination of the terms listed in Table A.1: ("Transitions Literature Term" OR "Transitions Literature Term Synonym") AND ("Multi-System Term" OR "Multi-System Term Synonym"). Synonyms were considered to encompass variations or alternative notations of the same term or concept. For instance, the search query might appear as follows: ("Multi-Level Perspective" OR "Multilevel Perspective" OR "MLP") AND ("Multi-Regime" OR "Multiple Regimes"). The databases used to identify this research were Google Scholar, Scopus, and Web of Science. These databases were also found to be the most frequently used by other researchers (Norris et al., 2008; Xiao and Watson, 2019). Google Scholar was used initially by searching the first five pages for all identified search-word combinations and adding all relevant papers based on the title and the abstract. After this, Scopus and Web-of-Science were used with the same keyword combinations to add additional relevant literature. In the first step, the general rationale was to include all seemingly relevant papers. After the first step, a total of 621 papers were added to the database. In the second step (screening), the duplicates were removed, and each paper was re-evaluated, mainly based on the abstract and introduction, after which 341 papers remained for further consideration.



Fig. 2. Technological Innovation System (TIS) framework (Bergek et al., 2015).



Fig. 3. Literature search and evaluation for inclusion.

The third step (eligibility) was divided into two parts. In the first part, we identified themes and sub-themes to assign the papers, allowing us to uncover the most relevant papers per theme. The two main themes were 'conceptual grounds' (further subdivided into main theoretical concepts) and 'multi-system transition (case) studies' (further subdivided per focal sector under study). The remaining papers' abstracts, introductions and conclusions were more carefully reviewed to decide their relevance to the research topic. A total of 216 papers remained and were distributed over both identified themes. In the second part, the entire list was sifted through once more, with the following criteria in mind: (i) only peer-reviewed articles were considered, and (ii) the quality and reliability of the journal were assessed by predominantly considering papers published in Q1 journals (at the time of the search between May 2022 - July 2022), with some exceptions of papers published in Q2 journals if deemed very relevant. Based on this quality check, the multi-system framework contributions of each of the papers were assumed to be relevant and were therefore considered in the current manuscript. A total of 70 papers remained. In the fourth and final step (inclusion), five studies were added by tracking down references within key documents found in an earlier stage (i.e., Backward snowballing). Therefore, after a careful review covering four steps, a total of 75 papers remained over the two main themes (67 in Q1 and 8 in Q2 journals).

To provide some details about the included papers, the 'conceptual grounds' theme consists of 35 papers. The first set of papers (23 out of 35) covers the main concepts within the sociotechnical transitions literature, including the Multi-Level Perspective (MLP), Strategic Niche Management (SNM) and Innovation Systems (IS). These papers were screened⁵ to form the first conceptualisation of the following aspects: (i) core concepts within the literature, (ii) specific definitions of core terms and (iii) conceptualisation of sectors and regimes. The other set of the papers within this theme (12 out of 35) covers conceptual grounds of multi-system analysis in sociotechnical transitions (focusing on the theory of economic sectors and the concept of sector coupling). These papers were screened to deepen the understanding of definitions and characteristics of multiple systems and sectors and how they transform. The papers in this theme serve as a background for the review presented in the current paper, and form the basis for the theoretical background described in Section 2. See Table B.1 in the appendix for a list of the papers within this theme.

Core to our literature review are the remaining 40 papers from the

 $^{^{5}}$ Screening in this context implies thoroughly reading the abstract, introduction, theoretical background and conclusions and skimming other chapters to find the above information (e.g., by searching for specific terms in the papers).



Fig. 4. Depiction of the publication dates of the reviewed papers. This shows that the number of papers has surged around 2017 onwards.

'multi-system transition (case) studies' theme, which cover both conceptual/theoretical and empirical (case) study papers and were subdivided based on the focal sector of study. These papers can be considered highly relevant according to the inclusion criteria. They are mainly built upon – and quite evenly distributed across – the abovedescribed two prominent (founding) theoretical frameworks within the field (van den Bergh et al., 2011; Markard et al., 2012), namely the Multi-Level Perspective (MLP) and Technological Innovation Systems (TIS), counting 19 MLP papers and 21 TIS papers. This distribution occurred despite our explicit search for papers that cover multiple frameworks within the transitions literature. The selected papers were read thoroughly and consequently were core to the results described in Section 4, reflecting the main research question discussed in the introduction. See Table B.2 within the appendix for a list of the papers in this theme.

4. Results

Within this section, we seek to answer the research question established in the introduction: "What insights do papers from the field of transition studies provide for the study of multi-system innovations from a sociotechnical perspective?" This will be done through a descriptive analysis of the reviewed papers, focusing on the Multi-Level Perspective (MLP) and the Technological Innovation Systems (TIS) papers from the 'multi-system transition (case) studies' theme as described in Section 3. These papers fit within the scope of this review since they attempt to extend the usual single-system focus to consider multi-system influences, either with a focus on two (inter-system) or more (multi-system) of the following energy-demanding sociotechnical systems: electricity, transport, industry, the built environment, and agriculture. For this, the papers usually adapt established frameworks (i.e., MLP and TIS) to include multi-system dynamics. As depicted in Fig. 4, based on our inclusion criteria, the selected papers explore multi-system interactions from around 2007 onwards (Geels, 2007; Raven and Verbong, 2007). Notably, in 2017, we observe a steep increase in the number of relevant multi-system papers.

While the papers have overlapping topics and results, we have subdivided them according to each study's core idea to structure the forthcoming analysis. The title of each sub-category is derived either from much-used terms in the clustered papers or the main concept that the papers jointly contribute. Each subsection will summarise the core concepts presented by its respective cluster of papers. Within these subsections, we focus the descriptive analysis on each cluster's main insights and contribution towards studying multiple systems (i.e., sectors or regimes) that moves away from the traditional focus on a single system. Moreover, the core concepts of each cluster are summarised in figures (see Figs. 5–12). The focus of this paper is on conceptual and analytical contributions and not on methodological suggestions.

4.1. Multi-Level Perspective (MLP) papers

We subdivided the MLP papers addressing multi-system transitions

across four main sub-categories; (i) Multi-regime Interactions, (ii) Coupled Value Chains, (iii) Transition Phases and (iv) Connected Transitions. Table B.2 in the appendix provides a list of papers per cluster. In the following, we will describe their contributions to the body of knowledge regarding multi-system interactions. Refer to Figs. 5–8 for a graphical summary of the core concepts of each cluster, respectively.

4.1.1. Multi-regime interactions

The sub-category comprising the 'multi-regime interactions' papers contain the biggest number of the reviewed MLP papers (9 out of 21). The papers generally focus on cases where more than one regime and the interplay between these regimes plays a role. These papers span back to 2007; between 2007 and 2008, five multi-regime papers were published, directly addressing the focus of MLP studies on single-regime interactions (e.g., Raven, 2007 or Verbong and Geels, 2007). Each paper included a multi-regime case study design, for example, by exploring the co-evolution of natural gas and electricity systems in the Netherlands (Raven and Verbong, 2007) or interaction patterns among different German utilities providing telecom, electricity, gas, and water (Konrad et al., 2008).

These early papers made two main contributions to multi-system studies. Firstly, they identified common interactions between regimes. For example, Raven and Verbong (2007) identified four types of interactions between regimes: (i) competition occurs when regimes start



Fig. 5. Aggregated depiction of the main frameworks used by the 'Multi-Regime Interactions' papers. The image is adapted from Sutherland et al. (2015) and graphically depicts two interacting regimes (i.e., cross-regime dy-namics) with multiple niches in development. Niches can also be clustered per sociotechnical system, as depicted by the dashed lines around the niches (e.g., Niche 1 and 2 are part of the Electricity system, and Niche 3 is part of the Agriculture system).



Fig. 6. Aggregated depiction of the main frameworks used by the 'Coupled Value Chains' papers. This is a schematic representation of a sociotechnical system's value chain. The image is adapted from McMeekin et al. (2019) while adding a schematic representation of different systems per part of the value chain as advocated in Meynard et al. (2017). Each part of the value chain consists of sub-systems that perform a different function within the sociotechnical system. Each sub-system is part of a bigger (overlapping) sociotechnical system interacts with multiple other sociotechnical systems. Multiple niche innovations can influence systems and parts of the value chain. Researchers can take different scopes in their study of sociotechnical complexity (Geels, 2018).



Fig. 7. Aggregated depiction of the main frameworks used by the 'Transition Phases' papers. The image merges the life-cycle model from Kanger (2021) with the MLP conceptualisation from Geels et al. (2017a) to depict transitions as interactions of regimes in different phases of their lifecycle (e.g., emergence, upscaling, consolidation, reorganisation, destabilisation and stagnation, and decline).

fulfilling similar functions; (ii) integration occurs when previously separated regimes overlap (e.g., through actor mergers); (iii) spill-over occurs when practices from one regime become adopted in another, and (iv) symbiosis occurs when the two regimes reap mutual benefits from each other's existence. Secondly, the papers highlight that these interactions evolve over time. For example, in Geels' (2007) exploration



Time (decades)

Fig. 8. Aggregated depiction of the main frameworks used by the 'Connected Transitions' papers. The image is a schematic representation of the theoretical framework depicted in Schot and Kanger (2018). Each (S-)curve represents a decades-long technology surge from the installation to the deployment phase. One overarching (path-dependent) Deep Transition may encompass multiple technology surges.

of the breakthrough of rock 'n' roll in the US, the relationship between the radio and recording regimes evolved from competition to symbiosis. Each paper concludes with a call for more explicit multi-regime analysis. Erlinghagen and Markard (2012) answer this call and reinforce these findings by showing that entrants from another sector (i.e., 'adjacents') can be important catalysts for sectoral transformation and should thus be analysed more systematically in transition studies.

Following up, Papachristos et al. (2013) performed a meta-analysis drawing on the cases from the previously discussed multi-regime studies. The paper contributes to multi-system analysis by introducing generalisable system interaction types. Some of the common patterns introduced: (i) the importance of old regimes for the development of a new system, (ii) the influence of specialised applications on new technologies, and (iii) the development and utilisation of complementary technologies and new skills. This paper reveals that interactions between systems may take on a reinforcing or disruptive pattern. A followup study by Sutherland et al. (2015) focusing on renewable electricity in the agriculture sector makes two main contributions to multi-system analysis. Firstly, it confirms that working across system boundaries (e. g., agriculture and electricity) is an important source of innovation and that anchoring into multiple regimes can increase the stability of niche development. Secondly, the study advocates researchers to specify the location of niches (e.g., the agriculture regime) instead of simply labelling niches 'outside' of the focal regime (e.g., the electricity regime). In their study, such approach allowed for clearer consideration of the effects of renewable energy transitions on the agriculture regime.

Finally, Rosenbloom (2019) emphasises caution towards the energy transition-related policy recommendations, referencing Raven (2007), since policies that target one system may directly or indirectly impact one or more other systems, modulating the course of societal development in unforeseen ways.

4.1.2. Coupled value chains

Three papers fall into the 'coupled value chains' sub-category. These papers generally emphasise the necessity of considering the whole value chain when analysing innovations in multiple sociotechnical systems. The authors usually refer to this as 'whole systems analysis'. For example, Meynard et al. (2017) provide a heuristic 'whole system' framework to organise the design of coupled agrifood innovations. Such a coupled design approach considers the potential synergies or antagonisms between agriculture and food systems involving their value chains (i.e., production, processing, distribution, and consumption). Geels (2018) similarly advocates that the usual analytical focus on 'singular disruption' should be complemented with greater attention to 'multiple innovations and system reconfiguration' (i.e., 'whole system' change focusing on coupled multi-system value chains). The paper points out that addressing system reconfiguration may require some reconceptualisation of the MLP to pay more attention to different kinds of change mechanisms, including (i) interactions between niche innovations, (ii) adoption of niche innovation(s) within existing systems, and (iii) interactions between multiple systems (Geels, 2018). Therefore, the author advocates incorporating insights from other social sciences and moving away from the Schumpeterian dichotomy in which new entrants overthrow locked-in incumbents. This is possible since MLP can be considered a Middle-Range Theory⁶ that conceptualises overall dynamic patterns in sociotechnical transitions, which can be adapted and expanded to encompass additional transition dynamics (Geels, 2011).

As a follow-up, in a case study on the UK electricity system, McMeekin et al. (2019) develop an approach for 'whole-system' analysis along with a framework that considers generation, distribution, and consumption, as well as the configuration of systems (e.g., electricity, agrifood, mobility). This study shows that interactions between different system configurations should be studied as an empirical matter since system architecture differs per system and can lead to varying implications for transition dynamics. Furthermore, this study finds that the UK's energy transition consists of multiple co-existing and interacting reconfiguration dynamics, which become more important as crosssystem linkages increase (e.g., between electricity and mobility). Therefore, it suggests to characterise the 'whole system transitions' as a (gradual) reconfiguration instead of the traditional conceptualisation of radical niche innovations substituting a prevailing regime, acknowledged as one of Geels and Schot's (2007) four possible transition pathways.

4.1.3. Transition phases

The 'transition phases' papers directly address the phases of the energy transition. They emphasise how developments in multiple sociotechnical domains can bring about 'momentum' that drives upscaling and that linkages between sociotechnical systems may stimulate deep and rapid decarbonisation (Geels et al., 2017a, 2017b; Kanger, 2021). One of the key themes in these papers concerns the timing, duration, and acceleration of transitions. We observe a renewed emphasis on conceptualisations of MLP, which subdivide the decades-long transition processes into four distinguishable phases (Geels, 2005; Geels et al., 2017a, 2017b; Kanger, 2021):

- Phase 1: Radical innovations emerge in (technological) niches;
- Phase 2: Innovations enter small market niches;
- Phase 3: Innovations break through and compete with the established regime;
- Phase 4: Regime substitution with the widespread adoption of new innovations.

Considering multi-system dynamics, Geels et al. (2017b) illuminate the importance of explicitly establishing the status of each sociotechnical system since they tend to be in different phases depending on the geographical location and the timing, which can influence the adoption of specific multi-system innovations like V2G. For example, in most Western countries, the progress of the transition in the electricity system (with wind and solar electricity diffusing rapidly) is greater than low-carbon transitions in passenger transport (where the petroleum-fuelled auto-mobility regime is still mostly deeply entrenched) (Geels et al., 2017b). In line with this, Kanger (2021) seeks to rethink the 'global' model of the MLP to include a more detailed understanding of how, when, and why transitions occur to advance research and stimulate effective interventions for accelerating transitions in energy and mobility systems.

4.1.4. Connected transitions

The papers in the 'connected transitions' sub-category argue that a structural solution to the current pressing global and interconnected (i. e., wicked) environmental problems requires radical transformations towards a sustainable society in the form of a Deep Transition: "a series of connected transitions in many sociotechnical systems [e.g., energy, mobility, food] towards a similar direction [e.g. sustainability]" (Schot and Kanger, 2018, p. 1045). These papers move the scope beyond 'system innovations' in a focal sector to resemble a Techno-Economic Paradigm (TEP) shift, which encompasses 40-60 year-long cyclical variations in economic growth (i.e., long waves) (Freeman and Perez, 1988; Mathews, 2013; Perez, 1983, 2003). Each wave evolves from small beginnings in specific sectors or regional areas and ends up encompassing entire economies and societies of certain countries, gradually diffusing to other countries. There have been five such waves since the beginning of the Industrial Revolution (Mathews, 2013). Radical innovations differ with respect to how pervasive their economic effects are (Schumpeter, 1939). While some radical innovations may influence only a specific sector or industry, others may affect a whole range of sectors or, in rare cases, the entire economy. To account for these dynamics in transitions, the papers in this sub-category merge the Techno-Economic Paradigm (TEP) framework with the Multi-Level Perspective (MLP) framework (Mathews, 2013; Schot and Kanger, 2018).

Two sequential papers by Schot and Kanger (2018) and Kanger and Schot (2019) emphasise that Deep Transitions are sociotechnical, transnational, multi-system transitions, which are even more difficult to understand and govern than transitions in individual sociotechnical systems. Techno-economic paradigms (TEPs) provide what they term a meta-regime: A coordinating mechanism generating interconnections between technologies and industries (i.e., a ruleset in multiple systems or regimes) (Schot and Kanger, 2018). They call the build-up of various sociotechnical systems taking place over the 19th and 20th centuries the First Deep Transition. According to the authors, the recent sustainability concerns stimulate the emergence of the Second Deep Transition. These papers together provide 12 overall propositions that constitute the macro-dynamics of the Deep Transitions framework (Kanger and Schot, 2019). Finally, Van der Vleuten (2019) contrasts these findings by highlighting that 'Deep Transitions' research should be further validated empirically, and assumptions should be submitted to the intersubjectivity of scholarly debate before taking definitions such as the 'First Deep Transition' for granted.

4.2. Technological Innovation System (TIS) papers

We subdivided the TIS papers addressing multi-system transitions into four main sub-categories; (i) Sectoral Change, (ii) Contextual Influence, (iii) Sectoral Configuration and (iv) Structural Tensions. Table B.2 in the appendix provides a list of papers per cluster. In the following, we will describe their contributions to the body of knowledge about multi-system interactions. Refer to Figs. 9–12 for a graphical summary of the core concepts of each cluster, respectively.

4.2.1. Sectoral change

The papers in the 'sectoral change' sub-category generally focus on the sectoral influence on TISs, for example, by introducing concepts from the Sectoral Systems of Innovations (SSI) field. To start, Dolata

⁶ Merton (1968:39) defined Middle-Range theories (MRT) as "theories that lie between the minor but necessary working hypotheses that evolve in abundance during day-to-day research and the all-inclusive systematic efforts to develop a unified theory that will explain all the observed uniformities of social behaviour, social organisation and social change". Geels (2011) emphasises that MLP is an MRT since it relates various concepts and uses empirical research to identify recurring patterns and generalisable lessons.

⁷ Geels and Schot (2007) develop a typology of four transition pathways: transformation, reconfiguration, technological substitution, and de-alignment and re-alignment. These pathways differ in combinations of timing and nature of Multi-level interactions.



Fig. 9. Aggregated depiction of the main frameworks used by the 'Sectoral Change' papers. The image depicts different TIS levels by adapting the Multi-Level IS framework from Cronin et al. (2022) and combines it with the heuristic framework for analysing distinct patterns of tech-based sectoral change from Dolata (2009). Here a sub-TIS can be a national-level TIS when the TIS is on the European level. The closed dots represent partners consisting of different actors (open circles). These actors can be part of a larger system, e.g., a Sectoral Innovation System (SIS). Key is the observation that technology-based sectoral change is characterised by (i) the transformative capacity of a new technology and (ii) the inherent adaptability of the sector (i.e., structures, institutions and actors). The integrated functional-structural TIS framework presented in Lamprinopoulou et al. (2014) can be used to analyse both micro-level and macro-level aspects.



Fig. 10. Aggregated depiction of the main frameworks used by the 'Contextual influence' papers. The image adapts the framework presented in Markard and Truffer (2008) and adds relationship types between niches from Markard and Hoffmann (2016). In essence, it depicts a TIS and key contextual elements, including all systems that it interacts with and different kinds of complementary and competing innovations that emerge within the focal TIS. The black double arrows indicate the sectors that the TIS interacts with, which is specified per niche; While one TIS can interact with multiple sectors, niches can also interact with multiple sectors (i.e., a multi-system niche). Since the TIS has a lifecycle, the configuration of TIS-context interaction can change over time and is distinctly different during expansion (maturation) and decline until the TIS disappears and is replaced by an adjacent (maturing) TIS.

(2009) produces a heuristic framework for analysing distinct patterns of technology-based sectoral change. The paper points to two key determinants of the likelihood that sectoral change arises through innovation: (i) the sector-specific transformative capacity of new



Fig. 11. Aggregated depiction of the main frameworks used by the 'Sectoral Configuration' papers, depicting the sectoral configuration of a TIS. The image adapts the framework presented in Stephan et al. (2017) on the left and adds a zoomed-in perspective on the right to represent sector-TIS interactions with a focus on structural overlaps, as depicted in Mäkitie et al. (2018). Each dashed circle on the left represents a different sector, and each dot is an actor. Each sector encompasses the actors producing similar outputs (e.g., components or sub-systems underlying the focal technology). As represented in the (zoomed-in) image on the right, between a sector and a TIS, there are structural overlaps (e.g., actors, technology, institutions, networks); Actors may share similar institutions and production techniques and are linked through sector-specific networks. One actor might be active in different sectors. The sectors are active in other TISs to varying degrees.



Fig. 12. Aggregated depiction of the main frameworks used by the 'Structural Tensions' papers, depicting a map of multi-technology interaction and associated sectors. The image adapts the framework presented in Andersen and Markard (2020) on the left, adding a zoomed-in perspective into the overlap between a Focal TIS and different sectors on the right (i.e., Sector-TIS interactions). This depicts TIS development cycles with structural tensions and development blocks that are influenced by complementary technologies as depicted in Haley (2018).

technologies themselves and (ii) the sectoral adaptability of socioeconomic systems confronted with the opportunities presented by new technologies. From these insights, the paper makes two main contributions to multi-system analysis. Firstly, it highlights the necessity to typify different sectors more precisely and determine distinct influencing factors that foster their adaptability (e.g., sociotechnical systems must show compatibility). Secondly, the framework highlights the importance of an inter-sectoral TIS view since the transformation of the 'focal' sector can be driven by technology developed 'outside' of that sector. This point was reiterated by Erlinghagen and Markard (2012), which showed that incumbents from adjacent sectors, such as IT or telecommunications, may drive innovation in the energy sector.

Building on this, Lamprinopoulou et al. (2014) draw up a multisystem structural-functional framework that combines the agrifood sectors to assess and compare the Agriculture Innovation Systems of Scotland and the Netherlands. Their framework includes transformational failures in addition to the established concepts of market failures and structural failures that were first introduced by Weber and Rohracher (2012). In this context, transformational failures represent a TIS's overall functioning and capability to renew itself and support major transitions as opposed to incremental improvement of the existing systems. From a multi-systems perspective, this allows for (traditional)

'structural innovation policies' - that address structural deficits in innovation systems - to be complemented by 'transformation-oriented innovation policies' which strategically focus on the transformation of whole systems of innovation, production and consumption (i.e., multisectoral). The stepwise framework introduced by Lamprinopoulou et al. (2014) indicates the need first to perform a micro-level failure analysis (i.e., functional analysis and structural-oriented analysis) and then a macro-level failure analysis (i.e., transformation-oriented analysis). Following up, Weber and Schaper-Rinkel (2017) introduce a multisector innovation foresight methodology to identify cross-cutting patterns of sectoral change and the effects on innovation policy at the European level. The paper shows that since innovation dynamics differ across sectors, exploring future perspectives on general innovation patterns needs to be rooted in the specific sectoral dynamics while allowing for the identification of (new) cross-sectoral patterns (e.g., to facilitate cross-sectoral 'Key Enabling Technologies'⁸). Based on the insights from five different sectors, the authors find a blurring of sectoral boundaries, along with sectoral and cross-sectoral integration and an increased need for governance of interactions between the sectors.

Finally, in a paper focused on transformative policies for sustainability transitions, Fagerberg (2018) highlights (i) a need to focus on supply and demand factors since both are deemed essential for success (i.e., a 'holistic perspective') and (ii) a need for innovative policy governance and instrumentation to account for policy coordination across different sectors to provide policymakers with tools to consider strategic, long-run societal (sustainability) goals.

4.2.2. Contextual influence

The 'contextual influence' sub-category constitutes contributions mainly (co-)authored by Markard. The papers make several key contributions to the multi-system analysis. Markard and Truffer (2008) and Wirth and Markard (2011) develop and use a merged MLP-TIS framework to encompass contextual influences, such as a TISs interaction with one or more sociotechnical regimes and several niches or application contexts. The studies validate that a TIS interacts with and is influenced by the focal sector (the sector it is mainly embedded in) and adjacent sectors (other related sectors); thus, many TISs are part of several sectors. Notably, there is a distinction between negative (competitive) and positive (complementary) sector-TIS interactions, highlighting the relevance of multiple competing vs complementary technologies within a focal sector. These papers point out whether and how an emerging technological field links up with the structures of existing sectors depends on 'complementarities' (i.e., other technologies that affect the focal technology).

Markard and Hoffmann (2016) and Markard et al. (2016) focus on the intricate dynamics of such complementarities. Their work demonstrates that different technologies can both compete and complement each other simultaneously, depending on whether the analytical focus is centred on the performance of the focal technology or the focal sector. For example, from a technology perspective, hydropower and wind power are competing electricity generation technologies, both competing for generation capacity in the electricity mix (i.e., technology-level competition). However, from a sectoral perspective, they are complementary in balancing electricity supply and demand (i. e., sector-level complementary); hydropower can serve as a stable base load to complement the variability of wind energy. Thus, it is important to specify the analytical focus when exploring complementarities. Furthermore, complementarities might change as technologies diffuse more widely or as the technology matures, and they may increase interconnection between different sectors (e.g., electric vehicles connect

⁸ The authors emphasise that Key Enabling Technologies (KETs) could provide the basis for innovation in a wide range of products and processes across all industrial sectors and could be important in solving Europe's major societal challenges.

electricity and transport sectors). In general, they point out that (i) the degree to which contextual elements matter for a focal technology should be determined empirically and (ii) that the relationships with context structures can change over time (e.g., as a technology matures and expands to novel sectors).

Following up, Markard (2018) points out that inter-sectoral relations become denser in the advanced stages of the TIS lifecycle, and Markard (2020) develops a TIS life cycle framework to encompass a TISs configuration evolution over time and distinguishes between four critical stages of TIS development: *formation, growth, maturity,* and *decline.* For sustainability transitions, such a life cycle perspective highlights two crucial aspects: (i) it requires the TIS to include both technology production and innovation (as opposed to earlier studies' emphasis on innovation), (ii) it also emphasises a TISs ending, considering the necessity of phasing out 'unsustainable' technologies.

4.2.3. Sectoral configuration

The 'sectoral configuration' sub-category generally demonstrates the importance of studying the interplay of downstream and upstream segments of focal technology value chains. To begin, Bergek et al. (2015) distinguish two ways in which multi-system interactions are established: (i) 'External links', which refer to one-way influence from a contextual element on a TIS, such as national institutions, politics, or sudden price shifts and (ii) 'structural couplings', which refer to a situation with shared components between a TIS and a sector (e.g., actors, institutions, networks and technology), possibly resulting in a two-way interaction where systems mutually influence each other. For example, a TIS's actors can be categorised under different sectoral subgroups, representing the overlaps between the TIS and various sectors. Mäkitie et al. (2018) focus on these structural couplings to extend and nuance the notion of inter-industry relationships (i.e., between an established sector and a nascent TIS) by considering four types of structural overlaps and by distinguishing between two phases of TIS development. The paper shows that structural overlaps can influence a TIS's functions. For example, structural overlaps can be important for creating positive relationships between sectors and a TIS since they facilitate different forms of resource redeployment from sectors to TISs.

Stephan et al. (2017) propose an explicit delineation of the 'sectoral configuration' of a TIS to classify the number and types of sectors that are linked via its value chain. The paper shows that sectoral configurations are particularly important for modern multi-component technologies (MCTs⁹), such as lithium-ion batteries since their underlying innovation and dynamics involve multiple sectors (e.g., chemical, electronics, and automobile). Furthermore, since the influence of sectors on TISs varies over time, the paper emphasises that the sectoral configuration deserves continued attention in future TIS analyses, especially when TISs centre around the MCTs. Malhotra et al. (2019) further integrate the sectoral configuration into TIS analysis by illustrating the generic sectoral configuration of a TIS's value chain and classifying the value chain segments of a generic MCT into seven categories from production equipment supply to end use. Each of these segments involves one or more sectors, and a single sector can be involved in more than one value chain segment. These papers demonstrate the value of explicitly analysing the sectoral configuration in the TIS analyses. For example, differences in sectoral characteristics (such as complexity and specificity of knowledge) can influence the patterns of inter-sectoral learning (e.g., purposive learning-by-interacting between sectors), which can lead to differences in knowledge development and diffusion.

Finally, Bento et al. (2021) build upon the finding that access to

⁹ Multi-Component Technologies (MCTs) and the former-mentioned Key Enabling Technologies (KET) can be categorised under the umbrella-term 'Multi-Purpose Technologies,' which includes technologies with multiple usesectors that affect different sectors in distinguishable ways.

resources available in the context depends on structural (sectoral) couplings to show that two factors form the basis for attracting different sectors to an emerging TIS (i.e., inter-sectoral relations): 'technological variety' and 'technological relatedness.' In studying the effect of resources on the inter-sectoral relations in TISs, the authors find that high technological capacity and sectoral proximity are not requirements for entry. This indicates opportunities for a greater variety of existing sectors to be mobilised than the literature usually considers (e.g., Boschma, 2017).

4.2.4. Structural tensions

The papers in the 'structural tensions' sub-category advocate for incorporating Industrial Dynamics to improve the knowledge about the micro-foundations of TIS growth. They do so in relation to the changing sectoral composition (i.e., emergence, growth, and decline of sectors) associated with the structural transformation of economies (Carlsson, 2016; Kuznets, 1971). Haley (2018) presents a TIS development cycle that combines the TIS framework's ability to illuminate policy failures with the idea that technological diffusion creates 'structural tensions' that introduce transformation pressure (i.e., mismatches between an innovation and its broader sectoral system) (based on Dahmén, 1988). Structural tensions can introduce both positive and negative transformation pressure. 'Positive transformation pressure' occurs when innovations create new opportunities for increasing the performance of the entire sector by combining complementary technologies, institutions, and firms (i.e., development block¹⁰) (Dahmén, 1988). 'Negative transformation pressure' occurs when innovations make the old ways of doing things obsolete or less effective, which can push economic sectors or actors out of the economy (i.e., creative destruction).

Andersen and Markard (2020) introduce a novel multi-technology map to complement existing frameworks as a 'mapping device' for multi-technology interaction. In a nutshell, their framework¹¹ includes different levels of the technology value chain. It distinguishes three types of multi-technology interaction: (i) technology interaction within the focal sector (i.e., intra-sector technology interaction), (ii) technology interaction between upstream (e.g., involved in sub-component production) and downstream (e.g., use-sectors to create products or services) parts of the value chain (i.e., inter-sector technology interaction), and (iii) technology interaction between focal and adjacent sector(s) (i. e., cross-sector technology interaction). Since inter-, intra- and crosssector forms of technology interactions often affect each other, they conceptualise these interlinked technologies and sectors as one larger technological system or development block (Hughes, 1983). Tensions can arise both internally and externally, which can be resolved by complementary innovations allowing the block to evolve further. As structural tensions occur and are resolved, the boundaries of the block can change (Taalbi, 2016).

5. Conclusion and discussion

This paper primarily sought to answer the following research question: "What insights do papers from the field of transition studies provide for the study of multi-system innovations from a sociotechnical perspective?" To answer this question, we have systematically reviewed transitions literature to identify analytical frameworks used to study multi-system (i.e., multi-sector and multi-regime) innovations from a sociotechnical perspective. After an iterative selection process, 75 papers were selected for this review. Out of these papers, 35 were instrumental in providing background information, while the remaining 40 research manuscripts were directly used to address the research question. These papers were mainly built upon and adapted two prominent (founding) theoretical frameworks within the field; the Multi-Level Perspective (MLP) and Technological Innovation Systems (TIS), counting 19 MLP papers and 21 TIS papers. The current review focused on the contributions of each of the papers in adapting the usual single-sector focus of the abovementioned frameworks to analyse multi-system influences.

This final Section is structured as follows. First, we provide the conclusions to the research question in Section 5.1 in the form of three key takeaways distilled from the reviewed studies, which constitute generalisable aspects that merit further consideration in future sociotechnical multi-system analyses. Subsequently, in Section 5.2, we discuss the implications of these key takeaways to lay the foundation for an integrative framework for multi-system analyses, which is absent in the existing literature. Lastly, in Section 5.3, we discuss the implications for future research and how such a framework can be further developed.

5.1. Conclusion: key takeaways for multi-system analysis

In this subsection we reflect on the results of the systematic review and the implications it has on multi-system analysis. In accordance with our research question, the results present a comprehensive overview of how multi-system analysis is conceptualised within transition studies. While all reviewed papers stem from transition studies, they exhibit substantial variations in terms of underlying assumptions, scope of examination, and empirical cases. For instance, in terms of underlying assumptions, there is an evident contrast between the grouped TIS and MLP studies due to their distinct core frameworks. Furthermore, variations also exist between the grouped studies within each conceptual approach. For example, regarding the scope of examination, 'connected transitions' papers seek to identify overarching patterns that transcend various systems, while 'multi-regime' studies adopt a narrower scope to concentrate on bilateral interdependencies between specific systems, such as agriculture and energy. Despite the variability between studies, three key aspects have emerged as pivotal in multi-system analyses, directly addressing our research question.

The first key aspect involves understanding the 'overarching directionality' (common trajectory) of multiple sociotechnical systems (i.e., 'meta-regime'), considering their common trajectory and mutual influence on each other, since the similar trajectories provide the context within which innovations develop (e.g., Schot and Kanger, 2018). For example, in light of the current energy transition, numerous individual systems are concurrently undergoing electrification. This aspect points to an ongoing debate in the literature about the appropriate scope of examination to adopt, such as macro-level (e.g., macro trends), systemlevel (e.g., systemic effects) or actor-level effects (e.g., cross-sector learning), and the need to explore innovation processes across different levels of analysis since these levels are interconnected, and mechanisms between them can influence each other, enhancing the explanatory power of system concepts (e.g., Cronin et al., 2022; Lamprinopoulou et al., 2014). It indicates the need to explore further how to reconcile and accommodate the study of multi-system innovation processes that cross different levels of analysis.

The second key aspect is the explicit analysis of each system's development phase during a transition. Sociotechnical systems can be in varying phases of development due to factors like timing, duration, location, and acceleration (e.g., Geels et al., 2017b; Markard, 2018; Kanger, 2021). For example, the electricity system transition towards sustainability is further along than the transport system transition in many Western countries. This impacts the implementation of innovations that intersect these two systems, such as V2G. Thus, devising

¹⁰ A development block describes how a set of core innovations generate structural tensions (Dahmén, 1988), 'technical imbalances' (Murmann and Frenken, 2006) or 'reverse salients' (Hughes, 1983) across related technologies and sectors that in turn generate pressure for change in other parts of the system.

¹¹ Underlying their framework is the presupposition that technologies are both combinatorial and recursive systems, implying that there is no specific scope or level of analysis (e.g., the electric vehicle, its engine or the battery) that is a priori better when studying technological change (Arthur, 2009; Sandén and Hillman, 2011).

interventions for accelerating transitions in two different systems or for supporting multi-system innovations requires explicit analysis of each system's specific phase and how, when, and why transitions occur in each system. In this context, reviewed studies highlight the importance of adopting a life-cycle approach, particularly as we enter a new phase of the energy transition, for example, with the necessity to phase out unsustainable technologies. Such an approach gains increasing significance as transitions progress, with denser multi-system relations (e.g., Bento et al., 2021; Stephan et al., 2017; Markard, 2018) and evolving relationships with complementary technologies in more advanced stages of a transition life-cycle (e.g., Markard and Hoffmann, 2016).

The third key aspect is the necessity of explicit examination of system configuration, which refers to the sociotechnical components and structures that make up the system(s) and how they shape system processes. This aspect can be further subdivided into five primary considerations. First, numerous papers emphasise the importance of considering the value chain associated with an innovation (i.e., whole system analysis) (e.g., Lamprinopoulou et al., 2014; Meynard et al., 2017). Second, they highlight the merit in specifying the 'sectoral configuration' such as the number and types of sectors linked to the value chain. This is especially important for multi-component technologies (e.g., Stephan et al., 2017; Malhotra et al., 2019). Third, the reviewed papers point out a need to perform intra- and inter-sector analysis. Intra-sector analysis helps typify different sectors since each has distinct characteristics, and to understand the differing repercussions that innovations may have on the structural and institutional arrangements of different sectors (e.g., Bento et al., 2021; Erlinghagen and Markard, 2012). Meanwhile, inter-sector analysis helps typify specific relationships and interactions between certain sectors, which can range from positive (complementary or symbiotic) to negative (competitive) patterns (e.g., Markard and Hoffmann, 2016; Mäkitie et al., 2018). Fourth, the analysis of complementarities (i.e., complementary innovations) is critical since transitions involve and depend upon them, for example, to create momentum (e.g., Geels, 2018; Meelen et al., 2021). The analytical focus on 'singular disruption' should be complemented with greater attention for multiple innovations and system reconfiguration (e.g., Geels, 2018). Fifth, the concept of structural tensions (such as the mismatches between an innovation and its wider sectoral system) or structural overlaps (i.e., facilitating flows of resources from sector to TIS, for example, for creating positive relationships) and the creation of development blocks of inter-related technologies can be introduced to understand innovation dynamics better (e.g., Haley, 2018; Andersen and Markard, 2020).

As a general concluding note, each reviewed paper re-iterated the need to consider multi-system aspects in future research, especially as the energy transition progresses and becomes increasingly complex. This point is particularly emphasised in relation to two facets. The first concerns innovation policy, which has the potential to affect a wide range of sociotechnical systems, potentially leading to far-reaching and unintended consequences. The second emphasises the necessity of accounting for multi-system interactions in macro-level transition pathways, beyond the mere influence of interactions within individual sociotechnical systems, such as niche-regime or incumbent-challenger dynamics. Reviewed papers make initial contributions towards conceptualising this added complexity in transition pathways (e.g., Geels, 2007; Papachristos et al., 2013; Rosenbloom, 2019).

5.2. Towards an integrative framework for multi-system analysis

The three key takeaways derived from this review can serve as a first step towards developing an integrative sociotechnical multi-system framework. Such a framework, which may take the form of an adapted MLP or TIS, or a synthesis of both (an MLP-TIS framework), represents a noteworthy gap in the existing literature. It has the potential to either be one comprehensive overarching framework or a collection of frameworks tailored to the specific objectives and emphasis of a multisystem study. While we advocate future studies to take the three key takeaways gleaned from this review into account, we recognise that their relevance depends on the specific focus of each study. In accordance with this perspective, researchers have the flexibility to adopt either a comprehensive approach that encompasses all three key aspects or a more specialised approach that combines and delves deeply into one or more of these aspects. Regardless, it is imperative for multi-system researchers to explicitly acknowledge the aspects they include within their scope and those they exclude, as this choice significantly influences their research outcomes.

As a first attempt, Fig. 13 presents a possible graphical depiction of a multi-system innovation framework, offering a clear and structured way to conduct sociotechnical multi-system analyses. The framework takes inspiration from the MLP to distinguish three distinct levels; The context level to account for the broader landscape-level influences that affect all sociotechnical systems and the associated innovations (depicted by vertical double arrows). The sociotechnical systems level, which entails a broader term to encompass sociotechnical regimes and sectors. And finally, the niche level, where the innovations reside and are protected. At the core of this framework is a hexagon symbolising the central multisystem innovation under study. This innovation resides at the intersection of two or more focal sociotechnical systems, emphasising their interconnectedness and mutual influences. There are inherent interactions between the niche- and sociotechnical system levels (depicted by vertical double arrows), while most of this interaction will take place through the multi-system innovation. Notably, the placement of this hexagon at the intersection of the system and niche levels does not represent an advanced developmental phase of the innovation, rather it indicates its central role in coupling the two systems. Future studies can zoom into the hexagon to distinguish distinct interaction (or coupling) types between the multi-system innovation and the sociotechnical systems.

To conduct a comprehensive multi-system analysis, the framework incorporates the key takeaways in the following ways. First, concerning system configuration, researchers can explicitly define and delineate the sociotechnical systems central to their study. This provides a clear focus on each system's role in the multi-system innovation, including their necessary interactions (depicted by the horizontal double arrow). Furthermore, within the niche level, complementary innovations impacting the central multi-system innovation can be depicted, which are sorted according to their respective systems, while acknowledging that some innovations may transcend multiple systems. Second, to consider distinct system phases, the entire system-specific boxes within the niche level can be placed at different heights, depending on each system's transition phase. Furthermore, a scale is integrated into the niche level of the framework to represent the developmental phase of innovations within the context of each system (denoted by the vertical dashed double arrow); denoting the furthest developed innovations at the top, near the system level and less-developed innovations at the bottom. Third, the framework takes into account the overarching directionality that can impact each focal system in similar ways, providing a broader context for understanding the multi-system innovation (denoted by the dashed box around the focal systems). This perspective helps researchers identify the conditions and influences shaping the multi-system innovation.

Fig 14 offers a practical illustration of the framework applied to the analysis of the multi-system Vehicle-to-Grid (V2G) innovation, which enables electric vehicles to not only be used for mobility but also to direct electricity back into the grid when needed. With regards to the system configuration, the framework indicates the two focal sociotechnical systems between which V2G anchors: electricity and transport. Furthermore, it depicts complementary and competing technologies, such as shared mobility services (pertaining to the transportation system) and distributed energy resources that are relevant to the electricity system. Regarding the overarching directionality, both systems align with the trend of electrification in response to the ongoing energy



Fig. 13. Possible graphical representation of a multi-system innovation framework.



Fig. 14. Possible graphical representation of a multi-system innovation framework applied to the Vehicle-to-Grid (V2G) innovation.

transition, as well as a shift towards increased demand-side engagement. Regarding the phase of each transition, there exists a discrepancy in the advancement of these systems; in many Western countries, the lowcarbon transition in the electricity system is further along than in the transport system. This is denoted by placing the system-specific box encompassing the electricity niches slightly higher than the box encompassing transport system niches. Furthermore, the niches within each box have been placed at different heights depending on their developmental phase.

5.3. Implications for future work

Within this subsection, we explore the implications for future work concerning both the further development of an integrative sociotechnical multi-system framework and the broader field of multi-system research. Here, we address three aspects that stand out as important in this context.

The first aspect pertains to the analytical focus and underscores the importance of niche analysis in multi-system research. Transitions literature highlights the central significance of multi-system innovations within niches, that anchor in multiple systems (e.g., V2G), in sociotechnical transitions; they can, for example, increase developmental stability and innovation resilience (Sutherland et al., 2015) or even accelerate transitions (Geels et al., 2017a). Given that multi-system innovations often lead to greater complexity and thus to new and difficult-to-estimate in-practice risks, niche protection becomes arguably more

pivotal (Büscher et al., 2020; Ornetzeder and Sinozic, 2020). This review shows that, while the results are focused on multi-system interactions, the (internal) niche dynamics, such as actor expectations, network building and learning processes, and how multi-system innovations influence and are influenced by multiple systems is understudied. Therefore, we recommend more in-depth focus on innovation analysis and internal niche dynamics in future multi-system studies. This could be done by using an adapted multi-system framework mentioned in Section 5.2, potentially serving as a validation of this new multi-system framework.

The second aspect relates to the discussion on the scope of examination mentioned in Section 5.1. Along with the typical focus of transition studies, the reviewed multi-system papers mainly focus on identifying macro-trends or systemic effects. Next to a need to explore further how to reconcile and accommodate the study of innovation processes that cross different levels of analysis, we argue that there is a lack of focus on actor-level effects, such as inter-sectoral actor relationships and cross-sector learning. Within this actor-level focus, we highlight the importance of including demand-side characteristics since those are pivotal for the success of low-carbon transitions involving millions of citizens with unique characteristics (e.g., user practices, beliefs, culture, and skills). Incorporating insights from social sciences (such as sociology and psychology) and introducing new methods (such as discrete choice modelling) into transition studies can help shed light on societal attitudes and consumer behaviour. Notably, the introduction of multi-system innovations can make the demand-side perspective more complex since various users, stemming from different sectors, may have different perspectives and interests regarding a certain innovation (which can serve distinct purposes depending on the focal sector) (Bettin, 2020; Bergek, 2019). Therefore, we recommend that future multi-system transition research and the associated multi-system framework further integrate demand-side characteristics.

The third aspect relates to a need to explore multiple methods used in future multi-system research. Aligned with the broader field, the reviewed multi-system papers are dominated by qualitative case-based research methods, aiming for more generic insights and explanations (Geels, 2007, 2010; Köhler et al., 2019). Considering the complex nature of multi-system analysis, it can be useful to explore complementary methodological approaches. Two promising, under-utilised methodological approaches for such complex analysis include processmethodological methods and simulation models.¹² Firstly, ontologically process-methodological methods (Garud and Gehman, 2012) fit well since transitions research is understood as a process of change with complex chains of causation, multiple actors, and dynamic framework conditions (Geels and Schot, 2007). Among others, specific approaches such as Event Sequence Analysis (ESA) allow for comparing multiple cases using pattern matching, where patterns constitute distinct transition pathways (Spekkink, 2013; Spekkink and Boons, 2016; Boons et al., 2014). Secondly, simulation models can be used to reduce complexity and identify essential factors and processes through various degrees of abstraction. 'Transition modelling' has recently become a growing research area (Timmermans and de Haan, 2008; Halbe et al., 2015; Köhler et al., 2018; Holtz et al., 2015). Köhler et al. (2018) provide a guideline towards choosing an appropriate modelling approach based on an analysis of six model types¹³ (Köhler et al., 2018). Notably, agentbased models (ABM) and system dynamics (SD) models or a combination of the two seem well-suited (Köhler et al., 2009, 2018; Walrave and Raven, 2016). In addition, proposals have been made to combine qualitative and quantitative approaches, in which models serve to check the internal consistency of narratives, which in turn inform models (Köhler et al., 2020; Turnheim et al., 2015; Trutnevyte et al., 2014;

Appendix A. Search strategy

Table A.1 Search terms.

Technological	Forecasting&	Social	Change 201	(2024)	123266

McDowall, 2014). Previously, there has been a lack of multi-system modelling research, which could be related to the associated complexity (Ramsebner et al., 2021). Thus, we recommend that future multi-system transition research implements these complementary methods, such as ESA and ABM, to capture the complexity of multi-system interactions.

The findings of this systematic literature review underscore the critical relevance of adopting a sociotechnical multi-system perspective in transition studies. The results not only shed light on the complexities of multi-system innovations but also offer invaluable insights on aspects to consider in future studies.

Funding source

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

CRediT authorship contribution statement

Jerico Bakhuis: Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Visualization, Writing – original draft. Linda Manon Kamp: Conceptualization, Supervision, Validation, Writing – review & editing. Natalia Barbour: Supervision, Writing – review & editing. Émile Jean Louis Chappin: Supervision, Writing – review & editing.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

Transitions Literature Terms	Multi-System Terms
Socio Technical Transition	Multi-System
Transition Theory	Multiple Sectors
Transition Management	Multiple Regimes
Transition Studies	Multi-Regimes
Sustainability Transition	Multi-Technology
Multi-level perspective	Multi-Purpose
Niche Development	Sector Coupling
Strategic Niche Management	Infrastructure
Innovation Theory	Converging Infrastructures
Innovation System	Convergence
Sectoral Innovation Systems	Vehicle to Grid
Regional Innovation System	Vehicle-to-X
Technical Innovation Systems	Car as a powerplant
System Dynamics	Sector
- Combination of MLP and Policy	- Inter-sector
- MLP and Quality	- Cross-sector
 MLP and Complexity 	- Multi-Sector
	Coupled Transition
	- Food, Water, Energy Nexus
	(continued on next page)

¹² See Köhler et al. (2019) for more details on state-of-the-art methodologies in Transition Studies.

¹³ The six models include: Eco-innovation (energy-economy models and Integrated Assessment Models); Evolutionary economics; Complex systems; Agent-Based Models (ABM); System Dynamics (SD); Socio-ecological systems (SES).

Table A.1 (continued)

Transitions Literature Terms	Multi-System Terms		
	- Food, water, Materials - Water-nexus - Agro-food		

Appendix B. List of papers

Table B.1

Conceptual grounds of sociotechnical transitions.

Main part	Cluster	Authors	Title	Journal Rank (Quartile) ^a
Main concepts within the transitions literature	General Studies	Markard et al. (2012)	Sustainability transitions: An emerging field of research and its prospects	Q1
		Köhler et al. (2019)	An agenda for sustainability transitions research: State of the art and future directions	Q1
		Loorbach et al. (2017)	Sustainability transitions research: transforming science and practice for societal change	Q1
	Multi-level Perspective	Geels (2002)	Technological transitions as evolutionary reconfiguration processes: a Multi-level perspective and a case-study	Q1
		Geels and Schot (2007)	Typology of sociotechnical transition pathways	Q1
		Geels (2011)	The Multi-level perspective on sustainability transitions: Responses to seven criticisms	Q1
		Geels (2010)	Ontologies, socio-technical transitions (to sustainability), and the Multi-level perspective	Q1
		Smith et al. (2010)	Innovation studies and sustainability transitions: The allure of the Multi-level perspective and its challenges	Q1
	Stratogia Niebo	Kaven (2007)	Niche accumulation and hybridisation strategies in transition processes towards a sustainable energy system: An assessment of differences and pitfalls Bogime chifte to quetrianability through processes of niche formation; the	QI
	Management	Schot and Geels	approach of strategic niche management Strategic niche management and stratinable innovation journeys: theory	Q2
		(2008) Turnheim and Geels	findings, research agenda, and policy Incumbent actors, guided search paths, and landmark projects in infra-system	01
		(2019)	transitions: Re-thinking Strategic Niche Management with a case study of French tramway diffusion (1971–2016).	¥.
		Van Eijck and Romijn (2009)	Prospects for Jatropha biofuels in Tanzania: An analysis with Strategic Niche Management	Q2
		Susur et al. (2019)	The emergence of regional industrial ecosystem niches: A conceptual framework and a case study.	Q1
	Transition Management	Rotmans et al. (2001)	More evolution than revolution: transition management in public policy	Q2
	Innovation Systems	Jacobsson and Bergek (2011)	Innovation system analyses and sustainability transitions: Contributions and suggestions for research	Q1
		Malerba (2002) Bergek et al.	Analyzing the functional dynamics of technological innovation systems: A	Q1 01
		(2008b)	scheme of analysis	¥-
		Suurs and Hekkert (2009)	Cumulative causation in the formation of a technological innovation system: The case of biofuels in the Netherlands	Q1
		Weber and Rohracher (2012)	Legitimizing research, technology and innovation policies for transformative change: Combining insights from innovation systems and Multi-level perspective	Q1
		Pigford et al. (2018)	in a comprenensive ranueres framework. Beyond AIS? Exploring an agricultural innovation ecosystems approach for niche design and development	Q1
	Combined Frameworks	Geels (2004)	From sectoral systems of innovation to socio-technical systems: Insights about dynamics and change from sociology and institutional theory	Q1
		Coenen and López (2010)	Comparing systems approaches to innovation and technological change for sustainable and competitive economies: an explorative study into conceptual	Q1
Multi-system analysis in	Theory on	Pavitt (1984)	Sectoral patterns of technical change: towards a taxonomy and a theory	Q1
sociotechnical transitions	(Economic) Sectors	Dosi (1988)	Sources, procedures, and microeconomic effects of innovation	Q1
		Carlsson and Stankiewicz (1991)	On the nature, function and composition of technological systems	QI
		Castellacci (2008)	Innovation and the competitiveness of industries: Comparing the mainstream and the evolutionary approaches	Q1
	The concept of Sector Coupling	Robinius et al. (2017a).	Linking the power and transport sectors—Part 1: The principle of sector coupling	Q2
		Robinius et al. (2017b)	Linking the power and transport sectors—Part 2: Modelling a sector coupling scenario for Germany	Q2
		Thellufsen and Lund (2017).	Cross-border versus cross-sector interconnectivity in renewable energy systems	Q1
		БГОШІ ет ал. (2018)	highly renewable European energy system	Ϋ́Υ

(continued on next page)

Table B.1 (continued)

Main part	Cluster	Authors	Title	Journal Rank (Quartile) ^a
		Arabzadeh et al. (2020)	Deep decarbonization of urban energy systems through renewable energy and sector-coupling flexibility strategies	Q1
		Fridgen et al. (2020)	A holistic view on sector coupling	Q1
		Ramsebner et al. (2021)	The sector coupling concept: A critical review	Q1
		Brear et al. (2020)	Sector coupling: supporting decarbonisation of the global energy system	Q2

^a Q: Defines the rank/quartile of the Journal (Q1 is occupied by the top 25 % of journals in the list; Q2 is occupied by journals in the 25 to 50 % group); C1: Total Citation in Scopus (date: between April 15 and May 1, 2022); C2 is total citation on Google Scholar (date: between April 15 and May 1, 2022), References are according to APA-style.

Table B.2

Multi-system transition (case) studies.

Main theory	Cluster	Authors	Title	Journal Rank (Quartile)
Multi-level Perspective	Multi-Regime Interactions	Geels (2007)	Analysing the breakthrough of rock 'n'roll (1930–1970) Multi-regime interaction and reconfiguration in the Multi-level perspective	Q1
		Raven (2007)	Co-evolution of waste and electricity regimes: multi-regime dynamics in the Netherlands (1969–2003)	Q1
		Verbong and Geels (2007)	The ongoing energy transition: lessons from a socio-technical, Multi-level analysis of the Dutch electricity system (1960–2004)	Q1
		Raven and Verbong (2007)	Multi-regime interactions in the Dutch energy sector: the case of combined heat and power technologies in the Netherlands 1970–2000	Q2
		Konrad et al. (2008)	Multi-regime dynamics in the analysis of sectoral transformation potentials: evidence from German utility sectors	Q1
		Erlinghagen and Markard (2012)	Smart grids and the transformation of the electricity sector: ICT firms as potential catalysts for sectoral change	Q1
		Papachristos et al. (2013)	System interactions in socio-technical transitions: Extending the Multi-level perspective	Q1
		Sutherland et al. (2015)	Conceptualising multi-regime interactions: The role of the agriculture sector in renewable energy transitions	Q1
		Rosenbloom (2019)	A clash of socio-technical systems: Exploring actor interactions around electrification and electricity trade in unfolding low-carbon pathways for Ontario	Q1
	Coupled Value Chains	Meynard et al. (2017) Geels (2018)	Designing coupled innovations for the sustainability transition of agrifood systems Disruption and low-carbon system transformation: Progress and new challenges in	Q1 01
		McMeekin et al. (2019)	socio-technical transitions research and the Multi-level Perspective Manning the winds of whole system reconfiguration: Analysing low-carbon	01
			transformations across production, distribution and consumption in the UK electricity system (1990–2016)	¥-
	Transition Phases	Geels et al. (2017a)	Sociotechnical transitions for deep decarbonization	01
		Geels et al. (2017b)	The Sociotechnical Dynamics of Low-Carbon Transitions	01
		Kanger (2021)	Rethinking the Multi-level Perspective for energy transitions: From regime life-cycle to explanatory typology of transition pathways	Q1
	Connected Transitions	Mathews (2013)	The renewable energies technology surge: A new techno-economic paradigm in the making?	Q2
		Schot and Kanger (2018)	Deep transitions: Emergence, acceleration, stabilization and directionality	Q1
		Kanger and Schot (2019)	Deep transitions: Theorizing the long-term patterns of socio-technical change	Q1
		Van der Vleuten (2019)	Radical change and deep transitions: Lessons from Europe's infrastructure transition 1815–2015	Q1
Technological Innovation Systems	Sectoral Change	Dolata (2009)	Technological innovations and sectoral change: Transformative capacity, adaptability, patterns of change: An analytical framework	Q1
		Lamprinopoulou et al. (2014)	Application of an integrated systemic framework for analysing agricultural innovation systems and informing innovation policies: Comparing the Dutch and Scottish agrifood sectors	Q1
		Weber and Schaper- Rinkel (2017)	European sectoral innovation foresight: Identifying emerging cross-sectoral patterns and policy issues	Q1
		Fagerberg (2018)	Mobilizing innovation for sustainability transitions: A comment on transformative innovation policy	Q1
		Cronin et al. (2022)	Multi-actor Horizon 2020 projects in agriculture, forestry and related sectors: A Multi- level Innovation System framework (MINOS) for identifying Multi-level system failures	Q1
	Contextual Influence	Markard and Truffer (2008)	Technological innovation systems and the Multi-level perspective: Towards an integrated framework	Q1
		Wirth and Markard (2011)	Context matters: How existing sectors and competing technologies affect the Swiss Bio-SNG innovation system.	Q1
		Markard and Hoffman (2016)	Analysis of complementarities: Framework and examples from the energy transition	Q1
		Markard et al. (2016)	Institutional dynamics and technology legitimacy–A framework and a case study on biogas technology	Q1
		Markard (2018)	The next phase of the energy transition and its implications for research and policy	Q1
			(contin	uued on next page)

Table B.2 (continued)

Main theory	Cluster	Authors	Title	Journal Rank (Quartile)
		Markard (2020)	The life cycle of technological innovation systems	Q1
	Sectoral Configuration	Bergek et al. (2015)	Technological innovation systems in contexts: Conceptualizing contextual structures and interaction dynamics	Q1
		Stephan et al. (2017)	The sectoral configuration of technological innovation systems: Patterns of knowledge development and diffusion in the lithium-ion battery technology in Japan	Q1
		Mäkitie et al. (2018)	Established sectors expediting clean technology industries? The Norwegian oil and gas sector's influence on offshore wind power	Q1
		Malhotra et al. (2019)	The role of inter-sectoral learning in knowledge development and diffusion: Case studies on three clean energy technologies	Q1
		Decourt (2019)	Weaknesses and drivers for power-to-X diffusion in Europe. Insights from technological innovation system analysis	Q1
		Bettin (2020)	Electricity infrastructure and innovation in the next phase of energy transition—amendments to the technology innovation system framework	Q1
		Bento et al. (2021)	Inter-sectoral relations to accelerate the formation of technological innovation systems: Determinants of actors' entry into marine renewable energy technologies	Q1
	Structural Tensions	Haley (2018)	Application to the case of transmission interconnections and renewable electricity in Nova Scotia. Canada	Q1
		Andersen et al. (2020)	The role of inter-sectoral dynamics in sustainability transitions: A comment on the transitions research agenda	Q1
		Andersen and Markard (2020)	Multi-technology interaction in socio-technical transitions: How recent dynamics in HVDC technology can inform transition theories	Q1

References

- Andersen, A.D., Markard, J., 2020. Multi-technology interaction in socio-technical transitions: how recent dynamics in HVDC technology can inform transition theories. Technol. Forecast. Soc. Chang. 151, 119802.
- Andersen, A.D., Steen, M., Mäkitie, T., Hanson, J., Thune, T.M., Soppe, B., 2020. The role of inter-sectoral dynamics in sustainability transitions: A comment on the transitions research agenda. Environ. Innov. Soc. Trans. 34, 348–351.
- Arabzadeh, V., Mikkola, J., Jasiūnas, J., Lund, P.D., 2020. Deep decarbonization of urban energy systems through renewable energy and sector-coupling flexibility strategies. J. Environ. Manag. 260, 110090.
- Arthur, W.B., 2009. The Nature of Technology: What It Is and How It Evolves. Simon and Schuster.
- Asheim, B.T., Isaksen, A., 1997. Location, agglomeration and innovation: towards regional innovation systems in Norway? Eur. Plan. Stud. 5 (3), 299–330.
- Bento, N., Fontes, M., Barbosa, J., 2021. Inter-sectoral relations to accelerate the formation of technological innovation systems: determinants of actors' entry into marine renewable energy technologies. Technol. Forecast. Soc. Chang. 173, 121136.
- Bergek, A., 2019. Technological innovation systems: a review of recent findings and suggestions for future research. In: Handbook of Sustainable Innovation, pp. 200–218.
- Bergek, A., Jacobsson, S., Hekkert, M., 2008a. Functions in Innovation Systems: A Framework for Analysing Energy System Dynamics and Identifying Goals for Systembuilding Activities by Entrepreneurs and Policy Makers.
- Bergek, A., Jacobsson, S., Carlsson, B., Lindmark, S., Rickne, A., 2008b. Analyzing the functional dynamics of technological innovation systems: a scheme of analysis. Res. Policy 37 (3), 407–429.
- Bergek, A., Hekkert, M., Jacobsson, S., Markard, J., Sandén, B., Truffer, B., 2015. Technological innovation systems in contexts: conceptualizing contextual structures and interaction dynamics. Environ. Innov. Soc. Trans. 16, 51–64.
- Bettin, S.S., 2020. Electricity infrastructure and innovation in the next phase of energy transition—amendments to the technology innovation system framework. Review of Evolutionary Political Economy 1 (3), 371–395.
- Bloess, A., 2019. Impacts of heat sector transformation on Germany's power system through increased use of power-to-heat. Appl. Energy 239, 560–580.
- Boons, F., Spekkink, W., Jiao, W., 2014. A process perspective on industrial symbiosis: theory, methodology, and application. J. Ind. Ecol. 18 (3), 341–355.
- Boschma, R., 2017. Relatedness as driver of regional diversification: a research agenda. Reg. Stud. 51 (3), 351–364.
- Böttger, D., Götz, M., Lehr, N., Kondziella, H., Bruckner, T., 2014. Potential of the powerto-heat technology in district heating grids in Germany. Energy Procedia 46, 246–253.
- Brear, M.J., Baldick, R., Cronshaw, I., Olofsson, M., 2020. Sector coupling: Supporting decarbonisation of the global energy system. Electr. J. 33 (9), 106832.
- Breschi, S., Malerba, F., 1997. Sectoral innovation systems: technological regimes, Schumpeterian dynamics, and spatial boundaries. Systems of Innovation: Technologies, Institutions and Organizations 1, 130–156.
- Brown, T., Schlachtberger, D., Kies, A., Schramm, S., Greiner, M., 2018. Synergies of sector coupling and transmission reinforcement in a cost-optimised, highly renewable European energy system. Energy 160, 720–739.
- Büscher, C., Scheer, D., Nabitz, L., 2020. Future converging infrastructures: assessing the consequences of increasing sector coupling and integration. TATuP-Zeitschrift für

Technikfolgenabschätzung in Theorie und Praxis/Journal for Technology Assessment in Theory and Practice 29 (2), 17–23.

- Carlsson, B., 2016. Industrial dynamics: a review of the literature 1990–2009. Ind. Innov. 23 (1). 1–61.
- Carlsson, B., Stankiewicz, R., 1991. On the nature, function and composition of technological systems. J. Evol. Econ. 1 (2), 93–118.
- Castellacci, F., 2008. Innovation and the competitiveness of industries: comparing the mainstream and the evolutionary approaches. Technol. Forecast. Soc. Chang. 75 (7), 984–1006.
- Child, M., Koskinen, O., Linnanen, L., Breyer, C., 2018. Sustainability guardrails for energy scenarios of the global energy transition. Renew. Sust. Energ. Rev. 91, 321–334.
- Coenen, L., López, F.J.D., 2010. Comparing systems approaches to innovation and technological change for sustainable and competitive economies: an explorative study into conceptual commonalities, differences and complementarities. J. Clean. Prod. 18 (12), 1149–1160.
- Cooke, P., Uranga, M.G., Etxebarria, G., 1997. Regional innovation systems: institutional and organisational dimensions. Res. Policy 26 (4–5), 475–491.
- Correlje, A., Verbong, G., 2004. The transition from coal to gas: radical change of the Dutch gas system. System Innovation and the Transition to Sustainability: Theory, Evidence and Policy 2004, 114–134.
- Cronin, E., Fieldsend, A., Rogge, E., Block, T., 2022. Multi-actor Horizon 2020 projects in agriculture, forestry and related sectors: a multi-level innovation system framework (MINOS) for identifying multi-level system failures. Agric. Syst. 196, 103349.
- Dahmén, E., 1988. 'Development blocks' in industrial economics. Scand. Econ. Hist. Rev. 36 (1), 3–14.
- Decourt, B., 2019. Weaknesses and drivers for power-to-X diffusion in Europe: Insights from technological innovation system analysis. Int. J. Hydrog. Energy 44 (33), 17411–17430.
- Dolata, U., 2009. Technological innovations and sectoral change: transformative capacity, adaptability, patterns of change: an analytical framework. Res. Policy 38 (6), 1066–1076.
- Dosi, G., 1988. Sources, procedures, and microeconomic effects of innovation. J. Econ. Lit. 1120–1171.
- Erlinghagen, S., Markard, J., 2012. Smart grids and the transformation of the electricity sector: ICT firms as potential catalysts for sectoral change. Energy Policy 51, 895–906.
- Fagerberg, J., 2018. Mobilizing innovation for sustainability transitions: a comment on transformative innovation policy. Res. Policy 47 (9), 1568–1576.

Freeman, C., 1987. Technical innovation, diffusion, and long cycles of economic

- development. In: The Long-wave Debate. Springer, Berlin, Heidelberg, pp. 295–309. Freeman, C., Perez, C., 1988. Structural crises of adjustment: business cycles. Technical Change and Economic Theory. Pinter, Londres.
- Fridgen, G., Keller, R., Körner, M.F., Schöpf, M., 2020. A holistic view on sector coupling. Energy Policy 147, 111913.
- Garud, R., Gehman, J., 2012. Metatheoretical perspectives on sustainability journeys: evolutionary, relational and durational. Res. Policy 41 (6), 980–995.
- Geels, F.W., 2002. Technological transitions as evolutionary reconfiguration processes: a multi-level perspective and a case-study. Res. Policy 31 (8–9), 1257–1274.
- Geels, F.W., 2004. From sectoral systems of innovation to sociotechnical systems: insights about dynamics and change from sociology and institutional theory. Res. Policy 33 (6–7), 897–920.

J. Bakhuis et al.

- Geels, F.W., 2005. Processes and patterns in transitions and system innovations: refining the co-evolutionary multi-level perspective. Technol. Forecast. Soc. Chang. 72 (6), 681–696.
- Geels, F.W., 2007. Analysing the breakthrough of rock 'n'roll (1930–1970) multi-regime interaction and reconfiguration in the multi-level perspective. Technol. Forecast. Soc. Chang. 74 (8), 1411–1431.
- Geels, F.W., 2010. Ontologies, socio-technical transitions (to sustainability), and the multi-level perspective. Res. Policy 39 (4), 495–510.
- Geels, F.W., 2011. The multi-level perspective on sustainability transitions: responses to seven criticisms. Environ. Innov. Soc. Trans. 1 (1), 24–40.
- Geels, F.W., 2018. Disruption and low-carbon system transformation: progress and new challenges in socio-technical transitions research and the multi-level perspective. Energy Res. Soc. Sci. 37, 224–231.
- Geels, F.W., Schot, J., 2007. Typology of sociotechnical transition pathways. Res. Policy 36 (3), 399–417.
- Geels, F., Sovacool, B., Schwanen, T., Sorrell, S., 2017a. Sociotechnical transitions for deep decarbonization. Science 357 (6357), 1242–1244.
- Geels, F.W., Sovacool, B.K., Schwanen, T., Sorrell, S., 2017b. The socio-technical dynamics of low-carbon transitions. Joule 1 (3), 463–479.
- Geels, F.W., Schwanen, T., Sorrell, S., Jenkins, K., Sovacool, B.K., 2018. Reducing energy demand through low carbon innovation: a sociotechnical transitions perspective and thirteen research debates. Energy Res. Soc. Sci. 40, 23–35.
- Halbe, J., Reusser, D.E., Holtz, G., Haasnoot, M., Stosius, A., Avenhaus, W., Kwakkel, J. H., 2015. Lessons for model use in transition research: a survey and comparison with other research areas. Environ. Innov. Soc. Trans. 15, 194–210.
- Haley, B., 2018. Integrating structural tensions into technological innovation systems analysis: application to the case of transmission interconnections and renewable electricity in Nova Scotia. Canada. Research Policy 47 (6), 1147–1160.
- Hekkert, M.P., Suurs, R.A., Negro, S.O., Kuhlmann, S., Smits, R.E., 2007. Functions of innovation systems: a new approach for analysing technological change. Technol. Forecast. Soc. Chang. 74 (4), 413–432.
- Holtz, G., Alkemade, F., De Haan, F., Köhler, J., Trutnevyte, E., Luthe, T., Ruutu, S., 2015. Prospects of modelling societal transitions: position paper of an emerging community. Environ. Innov. Soc. Trans. 17, 41–58.
- Hoogma, R., Kemp, R., Schot, J., Truffer, B., 2002. Experimenting for Sustainable Transport. Taylor & Francis.
- Hughes, T.P., 1983. Networks of Power: Electrification in Western Society, 1880–1930. JHU press.
- Jacobsson, S., Bergek, A., 2011. Innovation system analyses and sustainability transitions: contributions and suggestions for research. Environ. Innov. Soc. Transit. 1 (1), 41–57.
- Kanger, L., 2021. Rethinking the multi-level perspective for energy transitions: from regime life-cycle to explanatory typology of transition pathways. Energy Res. Soc. Sci. 71, 101829.
- Kanger, L., Schot, J., 2019. Deep transitions: theorizing the long-term patterns of sociotechnical change. Environ. Innov. Soc. Trans. 32, 7–21.
- Kemp, R., Schot, J., Hoogma, R., 1998. Regime shifts to sustainability through processes of niche formation: the approach of strategic niche management. Tech. Anal. Strat. Manag. 10 (2), 175–198.
- Köhler, J., Whitmarsh, L., Nykvist, B., Schilperoord, M., Bergman, N., Haxeltine, A., 2009. A transitions model for sustainable mobility. Ecol. Econ. 68 (12), 2985–2995.
- Köhler, J., De Haan, F., Holtz, G., Kubeczko, K., Moallemi, E.A., Papachristos, G., Chappin, E., 2018. Modelling Sustainability Transitions: An Assessment of Approaches and Challenges.
- Köhler, J., Geels, F.W., Kern, F., Markard, J., Onsongo, E., Wieczorek, A., Wells, P., 2019. An agenda for sustainability transitions research: state of the art and future directions. Environ. Innov. Soc. Trans. 31, 1–32.
- Köhler, J., Turnheim, B., Hodson, M., 2020. Low carbon transitions pathways in mobility: applying the MLP in a combined case study and simulation bridging analysis of passenger transport in the Netherlands. Technol. Forecast. Soc. Chang 151, 119314.
- Konrad, K., Truffer, B., Voß, J.P., 2008. Multi-regime dynamics in the analysis of sectoral transformation potentials: evidence from German utility sectors. J. Clean. Prod. 16 (11), 1190–1202.
- Kuznets, S., 1971. Economic Growth of Nations: Total Output and Production Structure. Harvard University Press.
- Lamprinopoulou, C., Renwick, A., Klerkx, L., Hermans, F., Roep, D., 2014. Application of an integrated systemic framework for analysing agricultural innovation systems and informing innovation policies: comparing the Dutch and Scottish agrifood sectors. Agric. Syst. 129, 40–54.
- Loorbach, D., Frantzeskaki, N., Avelino, F., 2017. Sustainability transitions research: transforming science and practice for societal change. Annu. Rev. Environ. Resour. 42, 599–626.
- Lund, H., Kempton, W., 2008. Integration of renewable energy into the transport and electricity sectors through V2G. Energy Policy 36 (9), 3578–3587.
- Lund, H., Möller, B., Mathiesen, B.V., Dyrelund, A., 2010. The role of district heating in future renewable energy systems. Energy 35 (3), 1381–1390.
- Lundvall, B.A., 1992. National Systems of Innovation: Towards a Theory of Innovation and Interactive Learning.
- MacInnis, D.J., 2011. A framework for conceptual contributions in marketing. J. Mark. 75 (4), 136–154.
- Mäkitie, T., Andersen, A.D., Hanson, J., Normann, H.E., Thune, T.M., 2018. Established sectors expediting clean technology industries? The Norwegian oil and gas sector's influence on offshore wind power. J. Clean. Prod. 177, 813–823.
- Malerba, F., 2002. Sectoral systems of innovation and production. Res. Policy 31 (2), 247–264.

- Malhotra, A., Schmidt, T.S., Huenteler, J., 2019. The role of inter-sectoral learning in knowledge development and diffusion: case studies on three clean energy technologies. Technol. Forecast. Soc. Chang. 146, 464–487.
- Markard, J., 2011. Transformation of infrastructures: sector characteristics and implications for fundamental change. J. Infrastruct. Syst. 17 (3), 107–117.
- Markard, J., 2018. The next phase of the energy transition and its implications for research and policy. Nat. Energy 3 (8), 628–633.
- Markard, J., 2020. The life cycle of technological innovation systems. Technol. Forecast. Soc. Chang. 153, 119407.
- Markard, J., Hoffmann, V.H., 2016. Analysis of complementarities: framework and examples from the energy transition. Technol. Forecast. Soc. Chang. 111, 63–75.
- Markard, J., Truffer, B., 2008. Technological innovation systems and the multi-level perspective: towards an integrated framework. Res. Policy 37 (4), 596–615.
- Markard, J., Raven, R., Truffer, B., 2012. Sustainability transitions: an emerging field of research and its prospects. Res. Policy 41 (6), 955–967.
- Markard, J., Suter, M., Ingold, K., 2016. Sociotechnical transitions and policy change–advocacy coalitions in Swiss energy policy. Environ. Innov. Soc. Trans. 18, 215–237.
- Mathews, J.A., 2013. The renewable energies technology surge: a new techno-economic paradigm in the making? Futures 46, 10–22.
- McDowall, W., 2014. Exploring possible transition pathways for hydrogen energy: a hybrid approach using socio-technical scenarios and energy system modelling. Futures 63, 1–14.
- McMeekin, A., Geels, F.W., Hodson, M., 2019. Mapping the winds of whole system reconfiguration: analysing low-carbon transformations across production, distribution and consumption in the UK electricity system (1990–2016). Res. Policy 48 (5), 1216–1231.
- Meelen, T., Doody, B., Schwanen, T., 2021. Vehicle-to-Grid in the UK fleet market: an analysis of upscaling potential in a changing environment. J. Clean. Prod. 290, 125203.
- Merton, R.K., 1968. Social Theory and Social Structure. Simon and Schuster.
- Meynard, J.M., Jeuffroy, M.H., Le Bail, M., Lefèvre, A., Magrini, M.B., Michon, C., 2017. Designing coupled innovations for the sustainability transition of agrifood systems. Agric. Syst. 157, 330–339.
- Mitchell, C., 2016. Momentum is increasing towards a flexible electricity system based on renewables. Nat. Energy 1 (2), 1–6.
- Murmann, J.P., Frenken, K., 2006. Toward a systematic framework for research on dominant designs, technological innovations, and industrial change. Res. Policy 35 (7), 925–952.
- Mwasilu, F., Justo, J.J., Kim, E.K., Do, T.D., Jung, J.W., 2014. Electric vehicles and smart grid interaction: a review on vehicle to grid and renewable energy sources integration. Renew. Sust. Energ. Rev. 34, 501–516.
- Norris, M., Oppenheim, C., Rowland, F., 2008. Finding open access articles using google, google scholar, oaister and opendoar. Online Inf. Rev. 32 (6), 709–715.
- Ornetzeder, M., Sinozic, T., 2020. Sector coupling of renewable energy in an experimental setting: findings from a smart energy pilot project in Austria. TATuP-Zeitschrift für Technikfolgenabschätzung in Theorie und Praxis/Journal for Technology Assessment in Theory and Practice 29 (2), 38–44.
- Papachristos, G., Sofianos, A., Adamides, E., 2013. System interactions in sociotechnical transitions: extending the multi-level perspective. Environ. Innov. Soc. Trans. 7, 53–69.
- Patel, P., Pavitt, K., 1994. National innovation systems: why they are important, and how they might be measured and compared. Econ. Innov. New Technol. 3 (1), 77–95.
- Pavitt, K., 1984. Sectoral patterns of technical change: towards a taxonomy and a theory. Res. Policy 13 (6), 343–373.
- Perez, C., 1983. Structural change and assimilation of new technologies in the economic and social systems. Futures 15 (5), 357–375.
- Perez, C., 2003. Technological Revolutions and Financial Capital. Edward Elgar Publishing.
- Pigford, A.A.E., Hickey, G.M., Klerkx, L., 2018. Beyond agricultural innovation systems? Exploring an agricultural innovation ecosystems approach for niche design and development in sustainability transitions. Agric. Syst. 164, 116–121.
- Ramsebner, J., Haas, R., Ajanovic, A., Wietschel, M., 2021. The sector coupling concept: a critical review. Wiley Interdisciplinary Reviews: Energy and Environment 10 (4), e396.
- Raven, R., 2007. Niche accumulation and hybridisation strategies in transition processes towards a sustainable energy system: an assessment of differences and pitfalls. Energy Policy 35 (4), 2390–2400.
- Raven, R., Verbong, G., 2007. Multi-regime interactions in the Dutch energy sector: the case of combined heat and power technologies in the Netherlands 1970–2000. Tech. Anal. Strat. Manag. 19 (4), 491–507.
- Rip, A., Kemp, R., 1998. Technological change. Human Choice and Climate Change 2 (2), 327–399.
- Robinius, M., Otto, A., Heuser, P., Welder, L., Syranidis, K., Ryberg, D.S., Stolten, D., 2017a. Linking the power and transport sectors—part 1: the principle of sector coupling. Energies 10 (7), 956.
- Robinius, M., Otto, A., Syranidis, K., Ryberg, D.S., Heuser, P., Welder, L., Stolten, D., 2017b. Linking the power and transport sectors—part 2: modelling a sector coupling scenario for Germany. Energies 10 (7), 957.
- Rogelj, J., Forster, P.M., Kriegler, E., Smith, C.J., Séférian, R., 2019. Estimating and tracking the remaining carbon budget for stringent climate targets. Nature 571 (7765), 335–342.
- Rosenbloom, D., 2019. A clash of socio-technical systems: exploring actor interactions around electrification and electricity trade in unfolding low-carbon pathways for Ontario. Energy Res. Soc. Sci. 49, 219–232.

J. Bakhuis et al.

Technological Forecasting& Social Change 201 (2024) 123266

- Rotmans, J., Kemp, R., Van Asselt, M., 2001. More evolution than revolution: transition management in public policy. Foresight 3 (1), 15–31.
- Sandén, B.A., Hillman, K.M., 2011. A framework for analysis of multi-mode interaction among technologies with examples from the history of alternative transport fuels in Sweden. Res. Policy 40 (3), 403–414.
- Schot, J., Geels, F.W., 2008. Strategic niche management and sustainable innovation journeys: theory, findings, research agenda, and policy. Tech. Anal. Strat. Manag. 20 (5), 537–554.
- Schot, J., Kanger, L., 2018. Deep transitions: emergence, acceleration, stabilization and directionality. Res. Policy 47 (6), 1045–1059.
- Schumpeter, J.A., 1939. Business Cycles, vol. 1. Mcgraw-hill, New York, pp. 161–174. Smith, A., Raven, R., 2012. What is protective space? Reconsidering niches in transitions to sustainability. Res. Policy 41 (6), 1025–1036.
- Smith, A., Voß, J.P., Grin, J., 2010. Innovation studies and sustainability transitions: the allure of the multi-level perspective and its challenges. Res. Policy 39 (4), 435–448. Snyder, H., 2019. Literature review as a research methodology: an overview and
- guidelines. J. Bus. Res. 104, 333–339.
- Sovacool, B., Noel, L., Axsen, J., Kempton, W., 2017. The future promise of vehicle-togrid (V2G) integration: a sociotechnical review and research agenda. Annu. Rev. Environ. Resour. 42, 377–406.
- Spekkink, W., 2013. Institutional capacity building for industrial symbiosis in the Canal Zone of Zeeland in the Netherlands: a process analysis. J. Clean. Prod. 52, 342–355.
- Spekkink, W.A., Boons, F.A., 2016. The emergence of collaborations. J. Public Adm. Res. Theory 26 (4), 613–630.
- Stephan, A., Schmidt, T.S., Bening, C.R., Hoffmann, V.H., 2017. The sectoral configuration of technological innovation systems: patterns of knowledge development and diffusion in the lithium-ion battery technology in Japan. Res. Policy 46 (4), 709–723.
- Susur, E., Hidalgo, A., Chiaroni, D., 2019. The emergence of regional industrial ecosystem niches: a conceptual framework and a case study. J. Clean. Prod. 208, 1642–1657.
- Sutherland, L.A., Peter, S., Zagata, L., 2015. Conceptualising multi-regime interactions: the role of the agriculture sector in renewable energy transitions. Res. Policy 44 (8), 1543–1554.
- Suurs, R.A., Hekkert, M.P., 2009. Cumulative causation in the formation of a technological innovation system: the case of biofuels in the Netherlands. Technol. Forecast. Soc. Chang. 76 (8), 1003–1020.
- Taalbi, J., 2016. Development Blocks and Structural Analysis. Routledge, New York, pp. 56–77.
- Thellufsen, J.Z., Lund, H., 2017. Cross-border versus cross-sector interconnectivity in renewable energy systems. Energy 124, 492–501.
- Timmermans, J., de Haan, H., 2008. Special issue on computational and mathematical approaches to societal transitions. Computational & Mathematical Organization Theory 14 (4), 263–265.
- Torraco, R.J., 2005. Writing integrative literature reviews: guidelines and examples. Hum. Resour. Dev. Rev. 4 (3), 356–367.
- Trutnevyte, E., Barton, J., O'Grady, Á., Ogunkunle, D., Pudjianto, D., Robertson, E., 2014. Linking a storyline with multiple models: a cross-scale study of the UK power system transition. Technol. Forecast. Soc. Chang. 89, 26–42.
- Turnheim, B., Geels, F.W., 2019. Incumbent actors, guided search paths, and landmark projects in infra-system transitions: re-thinking strategic niche management with a case study of French tramway diffusion (1971–2016). Res. Policy 48 (6), 1412–1428.
- Turnheim, B., Berkhout, F., Geels, F., Hof, A., McMeekin, A., Nykvist, B., van Vuuren, D., 2015. Evaluating sustainability transitions pathways: bridging analytical approaches to address governance challenges. Glob. Environ. Chang. 35, 239–253.
- Turnheim, B., Wesseling, J., Truffer, B., Rohracher, H., Carvalho, L., Binder, C., 2018. Challenges ahead: Understanding, assessing, anticipating and governing foreseeable societal tensions to support accelerated low-carbon transitions in Europe. In: Advancing Energy Policy. Palgrave Pivot, Cham, pp. 145–161.

UNFCCC, 2015. The Paris Agreement.

Van den Bergh, J.C., Truffer, B., Kallis, G., 2011. Environmental innovation and societal transitions: introduction and overview. Environ. Innov. Soc. Trans. 1 (1), 1–23.

- Van der Laak, W.W.M., Raven, R.P.J.M., Verbong, G.P.J., 2007. Strategic niche management for biofuels: analysing past experiments for developing new biofuel policies. Energy Policy 35 (6), 3213–3225.
- Van der Vleuten, E., 2019. Radical change and deep transitions: lessons from Europe's infrastructure transition 1815–2015. Environ. Innov. Soc. Trans. 32, 22–32.
- Van Eijck, J., Romijn, H., 2009. Prospects for Jatropha biofuels in Tanzania: an analysis with strategic niche management. In: Sectoral Systems of Innovation and Production in Developing Countries. Edward Elgar Publishing.
- Verbong, G., Geels, F., 2007. The ongoing energy transition: lessons from a sociotechnical, multi-level analysis of the Dutch electricity system (1960–2004). Energy Policy 35 (2), 1025–1037.
- Walrave, B., Raven, R., 2016. Modelling the dynamics of technological innovation systems. Res. Policy 45 (9), 1833–1844.
- Weber, K.M., Rohracher, H., 2012. Legitimizing research, technology and innovation policies for transformative change: combining insights from innovation systems and multi-level perspective in a comprehensive 'failures' framework. Res. Policy 41 (6), 1037–1047.
- Weber, K.M., Schaper-Rinkel, P., 2017. European sectoral innovation foresight: identifying emerging cross-sectoral patterns and policy issues. Technol. Forecast. Soc. Chang. 115, 240–250.
- Weber, K.M., Truffer, B., 2017. Moving innovation systems research to the next level: towards an integrative agenda. Oxf. Rev. Econ. Policy 33 (1), 101–121.
- Wee, B.V., Banister, D., 2016. How to write a literature review paper? Transp. Rev. 36 (2), 278–288.
- Wirth, S., Markard, J., 2011. Context matters: how existing sectors and competing technologies affect the prospects of the Swiss Bio-SNG innovation system. Technol. Forecast. Soc. Chang, 78 (4), 635–649.
- Xiao, Y., Watson, M., 2019. Guidance on conducting a systematic literature review. J. Plan. Educ. Res. 39 (1), 93–112.

Jerico Bakhuis is a PhD Candidate within the faculty Technology, Policy and Management of Delft University of Technology, the Netherlands. Prior to starting his PhD, Jerico finished two MSc programmes, the first in Global Business & Sustainability from Rotterdam School of Management and the second in Sustainable Energy Technology (SET) from Delft University of Technology. His field of research addresses the increasing need for efficient implementation of intermittent renewable energy from a socio-technical perspective with a focus on the role that demand-side innovations play in this.

Linda M. Kamp is an Assistant Professor in sustainable innovations and transitions within the faculty Technology, Policy and Management of Delft University of Technology, the Netherlands. Linda's research focuses on the implementation of sustainable energy technologies in developed and developing countries, and design of business strategies and business models to better facilitate this implementation. She has co-organised conference tracks and a special issue on this topic and she has published articles in leading international journals. Within the TU Delft MSc Sustainable Energy Technology (SET) she leads the Economics and Society graduation profile.

Natalia Barbour is an Assistant Professor at the University of Central Florida (UCF) where she studies transportation safety, travel behaviour, mode switch, and mobility adoption. She holds a doctoral degree in civil engineering (transportation) from the University of South Florida. Before joining UCF, Natalia was an Assistant Professor at Delft University of Technology and completed a postdoctoral training at Massachusetts Institute of Technology.

Emile Chappin is an Associate Professor at within the faculty Technology, Policy and Management of Delft University of Technology, head of the section Energy & Industry and co-director of the TPM Energy Transition Lab. Emile's research focuses on the role of simulation models and games for the energy transition, energy system analysis, and energy policy. In his research, Emile connects developments in modelling methodology to an understanding of complex socio-technical energy systems in order to better understand how to model, analyse and design (parts) of the energy transition.